Ther Adv Neurol Disord

2023, Vol. 16: 1–19 DOI: 10.1177/ 17562864231193530

© The Author(s), 2023. Article reuse guidelines: sagepub.com/journalspermissions

Correspondence to: **Eugen Trinka** Department of Neurology, Christian-Doppler University Hospital, Paracelsus Medical University, Centre for Cognitive Neuroscience, Member of the European Reference Centre EpiCARE, Ignaz Harrerstrasse 79, Salzburg A-5020, Austria

Neuroscience Institute, Christian-Doppler University Hospital, Paracelsus Medical University, Centre for Cognitive Neuroscience, Salzburg, Austria Institute of Public Health,

Medical Decision-Making and HTA, UMIT – Private University for Health Sciences, Medical Informatics and Technology, Hall in Tyrol, Austria

#### e.trinka@salk.at

Rodrigo Rocamora Hospital del Mar Medical Research Institute, Barcelona Spain

Epilepsy Monitoring Unit, Department of Neurology, Hospital del Mar, Barcelona, Member of the European Reference Centre EpiCARE, Spain

Faculty of Health and Life Sciences, Universitat Pompeu Fabra, Barcelona, Spain

## Eugen Trinka<sup>(D)</sup>, Rodrigo Rocamora, João Chaves, Mathias J. Koepp, Stephan Rüegg, Martin Holtkamp, Joana Moreira, Miguel M. Fonseca, Guillermo Castilla-Fernández and Fábio Ikedo

Lipid profile with eslicarbazepine acetate

adult patients with newly diagnosed focal

seizures: post hoc analysis of a phase III

and carbamazepine monotherapy in

trial and open-label extension study

### Abstract

**Background:** Antiseizure medications can have negative effects on plasma lipid levels. **Objectives:** To evaluate plasma lipid changes in patients with newly diagnosed focal epilepsy treated with eslicarbazepine acetate (ESL) or controlled-release carbamazepine (CBZ-CR) monotherapy during a phase III, randomized, double-blind (DB) trial and 2 years of ESL treatment in an open-label extension (OLE).

Design: Post hoc analysis of a phase III trial and OLE study.

**Methods:** Proportions of patients with elevated levels of total cholesterol and low-density lipoprotein (LDL) cholesterol were assessed at DB baseline, OLE baseline (last visit of DB trial), and end of OLE.

**Results:** A total of 184 patients received ESL monotherapy during the OLE: 96 received ESL monotherapy in the DB trial and 88 patients received CBZ-CR monotherapy. The proportions of patients with elevated total cholesterol and LDL cholesterol increased significantly during the DB trial in those treated with CBZ-CR monotherapy [total cholesterol, +14.9% (p < 0.001); LDL cholesterol, +11.5% (p = 0.012)] but decreased significantly after switching to ESL monotherapy in the OLE [total cholesterol, -15.3% (p = 0.008); LDL cholesterol, -11.1% (p = 0.021)]. No significant changes were observed in those treated with ESL monotherapy during the DB trial and OLE. At the end of the DB trial, between-group differences (ESL-CBZ-CR) in the proportions of patients with elevated total and LDL cholesterol were -13.6% (p = 0.037) and -12.3% (p = 0.061), respectively; at the end of the OLE, these between-group differences were -6.0% (p = 0.360) and -0.6% (p = 1.000), respectively.

**Conclusion:** A lower proportion of patients with newly diagnosed focal epilepsy had increased levels of total and LDL cholesterol, compared to baseline, following monotherapy with ESL *versus* CBZ-CR; after switching from CBZ-CR to ESL, the proportions of patients with increased levels decreased significantly.

**Registration:** ClinicalTrials.gov NCT01162460/NCT02484001; EudraCT 2009-011135-13/2015-001243-36.

journals.sagepub.com/home/tan



Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-NonCommercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission provided the original work is attributed as specified on the Sage and Open Access pages (https://us.sagepub.com/en-us/nam/open-access-at-sage).

#### João Chaves Department of Neurology, Hospital Santo António, Centro Hospitalar Universitário do Porto, Porto, Portugal

Mathias J. Koepp Department of Clinical and Experimental Epilepsy, UCL Queen Square Institute of Neurology, London, UK Chalfont Centre for

Epilepsy, Chalfont St Peter, UK

Stephan Rüegg Department of Neurology, Hospital of the University of Basel and University of Basel, Basel, Switzerland

Martin Holtkamp Department of Neurology, Epilepsy-Center Berlin-Brandenburg, Charité - Universitätsmedizin Berlin, Berlin, Germany

Joana Moreira Miguel M. Fonseca Fábio Ikedo Bial – Portela & Cª, S.A., Coronado, Portugal

Guillermo Castilla-Fernández Bial R&D Investments, S.A., Coronado, Portugal

#### Plain language summary

# The impact of treatment with either eslicarbazepine acetate or controlled-release carbamazepine on cholesterol levels in patients with newly diagnosed focal epilepsy

Patients with epilepsy have an increased risk of having cardiovascular and cerebrovascular diseases (e.g., myocardial infarction and stroke). Treatment with antiseizure medications can have a negative effect on blood cholesterol levels [such as total cholesterol and low-density lipoprotein (LDL) cholesterol, which can further increase the risk of cardiovascular and cerebrovascular diseases. We examined the impact of monotherapy treatment (i.e., treatment with only one antiseizure medication) using either eslicarbazepine acetate (ESL) or a controlledrelease formulation of carbamazepine (CBZ-CR) in 184 patients with newly diagnosed focal epilepsy (ESL, 96 patients; CBZ-CR, 88 patients). Patients received monotherapy with ESL or CBZ-CR for approximately 1 year in a phase III clinical trial. After this, the patients could continue into a 2-year extension study during which they all received monotherapy with ESL. We assessed the proportions of patients with elevated levels of total cholesterol and LDL cholesterol at the beginning and end of the phase III trial, and at the end of the extension study. At the beginning of the phase III trial, the proportions of patients with elevated total cholesterol and elevated LDL cholesterol were similar between treatment groups. During the phase III trial, the proportions of patients with elevated total cholesterol and elevated LDL cholesterol increased in those treated with CBZ-CR monotherapy (total cholesterol, +14.9%; LDL cholesterol, +11.5%) but decreased after switching to ESL monotherapy in the extension study (total cholesterol, -15.3%; LDL cholesterol, -11.1%). By contrast, the proportions of patients with elevated levels of total cholesterol and LDL cholesterol remained relatively stable in those treated with ESL monotherapy during the phase III trial and extension study. These findings indicate that ESL monotherapy may be an appropriate treatment option for patients with newly diagnosed focal epilepsy who either already have, or who are at risk of developing, high levels of cholesterol, since this may reduce their likelihood of having cardiovascular and cerebrovascular diseases.

*Keywords:* cardiovascular risk, cholesterol, focal epilepsy, focal seizures, hypercholesterolemia, lipid parameters, triglycerides

Received: 20 December 2022; revised manuscript accepted: 20 July 2023.

#### Introduction

Eslicarbazepine acetate (ESL) is a once-daily (QD) antiseizure medication (ASM) that is approved in Europe and the USA for the treatment of focal seizures as monotherapy or adjunctive therapy.<sup>1,2</sup> ESL was approved by the European Medicines Agency as monotherapy based on a phase III, randomized, double-blind (DB), active-controlled, non-inferiority study (Study BIA-2093-311), which demonstrated that monotherapy treatment with ESL QD was non-inferior to monotherapy treatment with

twice-daily (BID) controlled-release carbamazepine (CBZ-CR).<sup>3</sup>

Patients with epilepsy have an increased risk of cardio- and cerebrovascular (CCV) diseases and ASMs have been shown to exert different effects on a range of CCV risk factors, including plasma lipid levels (total cholesterol, low-density lipoprotein (LDL) cholesterol, high-density lipoprotein (HDL) cholesterol and triglycerides).<sup>4-9</sup> These effects are thought to arise from the variable enzyme-inducing properties of certain ASMs,

#### Study population

which increase a number of serological vascular risk factors, particularly cholesterol.<sup>10–13</sup> The use of enzyme-inducing ASMs is associated with an increased risk of CCV diseases over the long term,<sup>14,15</sup> while the use of enzyme-inhibiting ASMs has been shown to increase the risk of type 2 diabetes over the long term.<sup>16</sup> Choice of ASM monotherapy in the post-stroke epilepsy setting has also been shown to affect the relative risks of subsequent cardiovascular and all-cause death.<sup>17,18</sup> Since epilepsy is a chronic condition, often requiring life-long ASM treatment, the impact of an ASM on CCV risk factors is an important consideration when choosing treatment in order to reduce the risk of stroke and myocardial infarction over the long term.<sup>10</sup>

As a potent enzyme inducer, carbamazepine (CBZ) can lead to changes in lipid parameters *via* effects on enzymes involved in endogenous metabolic pathways.<sup>19</sup> In comparison, ESL is a less potent enzyme inducer than CBZ, and patients who have switched from CBZ to ESL have experienced improvements in lipid parameters.<sup>6,19–26</sup> This *post hoc* analysis evaluated changes in plasma lipid parameters for patients treated with ESL or CBZ-CR monotherapy during a randomized DB trial comparing CBZ-CR with ESL for newly diagnosed epilepsy,<sup>3</sup> and during the following 2 consecutive years of ESL treatment in an openlabel extension (OLE) study.<sup>27</sup>

#### Methods

#### Study design

Study BIA-2093-311 was a phase III, randomized, DB, non-inferiority trial, in which adult patients with newly diagnosed focal epilepsy received monotherapy with ESL QD (800, 1200 or 1600 mg/day) or CBZ-CR BID (400, 800 or 1200 mg/day) over a 26-week evaluation period, followed by a 26-week maintenance period (Figure 1).<sup>3</sup> Patients completing the DB trial (i.e., who remained seizure-free for  $\geq 6$  months at the last evaluated dose) were eligible to enter a 2-year OLE study, during which all patients received flexibly dosed ESL (800-1600 mg/day) (Figure 1).<sup>27</sup> The switching strategy used for transitioning patients from CBZ-CR to ESL at the start of the OLE study has been described previously (Trinka et al.<sup>27</sup>; Supplemental Material). According to the investigator's opinion, concomitant ASMs could be added during the OLE study, if required.<sup>27</sup>

The DB trial included adult patients ( $\geq 18$  years) with newly diagnosed epilepsy, providing they had at least two focal seizures (with or without focal to bilateral tonic-clonic seizures) within 12 months previous to screening and at least one seizure during the previous 3 months.<sup>3</sup> Patients were also required to have: (i) had an electroencephalogram (EEG) and brain computerized axial tomography or magnetic resonance imaging (to exclude a progressive neurological lesion) in the previous 12 months; (ii) provided written informed consent; and (iii) demonstrated cooperation and willingness to complete the study.<sup>3</sup> Key exclusion criteria included: (i) indicators of generalized genetic epilepsy (based on clinical history, specific seizure types and EEG); (ii) history of a clinically significant or currently uncontrolled medical condition that would interfere with participation in the study (e.g. renal disorder, hepatic disorder, uncontrolled hypertension, psychiatric illness); and (iii) previous treatment with CBZ or oxcarbazepine.<sup>3</sup> For inclusion in the OLE study, patients had to participate in the DB trial, with treatment ongoing at the time of unblinding treatment allocation.27 Key exclusion criteria for the OLE study included: (i) the presence of any major protocol violation during the initial DB trial that could impact compliance in the OLE study; (ii) suicidal risk (based on the investigator's opinion and the Columbia Suicide-Severity Rating Scale); (iii) occurrence of an adverse event in the initial trial indicating suspected presence of atrioventricular block (second degree or above) or contraindicative to further participation (investigator's judgment); (iv) events of alcohol/drug/medication abuse during the initial trial; (v) presence of clinically relevant laboratory abnormalities (e.g. sodium <125 mmol/L); and (vi) pregnancy or lactation.27

Patients who required additional ASM treatment during the OLE study were excluded from the analysis described below in order to evaluate a true monotherapy population.

#### Study assessments

Plasma levels of total cholesterol, LDL cholesterol, HDL cholesterol and triglycerides were assessed at baseline of the DB trial (DB baseline), at baseline of the OLE study (OLE baseline, which corresponds to the end of the DB trial),



**Figure 1.** Study design. Figure reproduced from Trinka *et al.*<sup>27</sup> with permission from Wiley. \*If seizures occurred during the evaluation period of the DB study, patients were assigned to the next dose level using 1-week titration period (CBZ-CR required titration, ESL did not) and 1-week stabilization period, followed by 26-week evaluation period as before.

<sup>†</sup>Patients who remained seizure-free for 26 weeks at any dose during the evaluation period entered the 26-week maintenance period. After the 26-week maintenance period, patients continued in an extension phase until the database lock (the database was locked after the last patient had completed the 26-week maintenance period). For all patients participating in the OLE study, the last extension phase visit of the DB study was also OLE visit 1 for the OLE study. <sup>¶</sup>Patients who received CBZ-CR during the DB trial transitioned to ESL at the start of the OLE study. BID, twice daily; CBZ-CR, controlled-release carbamazepine; DB, double-blind; ESL, eslicarbazepine acetate; OLE, openlabel extension; QD, once daily.

and at the end of study (EOS) in the OLE (OLE EOS).

#### Statistical analysis

Due to high variability in the data, it was not possible to conduct meaningful analyses of mean change in lipid parameters. Instead, an approach more aligned with clinical practice was adopted, in which the proportions of patients with high/low plasma levels of lipid parameters (according to reference laboratory values) were assessed to provide a clinically meaningful understanding of changes in lipids parameters over time. The proportions of patients with high/low plasma levels of lipid parameters (Lipid Reference Laboratory: LKF – Laboratorium für Klinische Forschung – POB 7134 - D-24171 Kiel, Germany) were calculated for the total monotherapy population, and for the subset of patients not treated concomitantly with statins [hydroxymethylglutaryl-coenzyme A (HMG-CoA) reductase inhibitors], since evidence has demonstrated that some ASMs, such as CBZ and phenytoin, significantly affect the lipid-lowering effects of statins.7,28 Plasma levels were considered high for total cholesterol if > 6.22 mmol/L and normal if  $\leq 6.22 \text{ mmol/L}$ . For LDL cholesterol, plasma levels were considered high if >4.14 mmol/L and normal if ≤4.14 mmol/L. For HDL cholesterol, plasma levels were considered high if  $> 2.07 \,\mathrm{mmol/L}$ , normal if  $\geq 1.04$  and  $\leq 2.07 \text{ mmol/L}$ , and low if



**Figure 2.** Patient disposition. Figure adapted from Trinka *et al.*<sup>27</sup> with permission from Wiley (a) Patients treated with ESL throughout the study (DB trial and OLE study). (b) Patients treated with CBZ-CR monotherapy in the DB trial and ESL in the OLE study. AE, adverse event; ASM, antiseizure medication; CBZ-CR, controlled-release carbamazepine; DB, double-blind; ESL, eslicarbazepine acetate; FAS, full analysis set; OLE, open-label extension; SAF, safety population.

<1.04 mmol/L. For triglycerides, plasma levels were considered high if > 1.70 mmol/L and normal if  $\leq 1.70 \text{ mmol/L}$ . Qualitative data were statistically evaluated using the McNemar test (for comparison between visits inside each treatment arm) and Fisher's test (for comparisons between treatment arms at each visit). Additional analyses were conducted to estimate the risk of developing high levels of total cholesterol or LDL cholesterol separately during the DB trial, and to assess the relative risk reduction for ESL versus CBZ-CR. These were assessed by analyzing cumulative risk (cumulative incidence), risk differences and relative risks [with 95% confidence interval (CI)] with their corresponding *p* values. No multiplicity adjustment of p values was conducted for any statistical testing due to the exploratory nature of the study. The standard alpha level of 0.05 was used, but statistically significant p values should be interpreted carefully and only within the context of this post hoc clinical evaluation.

#### Results

#### Study population

Of 206 randomized patients, 184 (89.3%) were treated with ESL monotherapy during the OLE study, of whom 96 had received ESL monotherapy in the DB trial ('ESL/ESL' treatment group) and 88 patients had received CBZ-CR monotherapy in the DB trial ('CBZ-CR/ESL' treatment group) (Figure 2). The remaining 22 randomized patients (10.7% of the randomized population) required concomitant ASM treatment during the OLE study and were therefore excluded from the analysis. The median (range) exposures to ESL and CBZ-CR during the DB trial were 1089 (675–1834) and 1165 (716–1904) days, respectively. During the OLE study, the median (range) exposures to ESL in the ESL/ESL and CBZ-CR/ESL treatment groups were 734 (39–769) and 730.5 (39–758) days, respectively.

The mean [standard deviation (SD)] age of the study population was 42.1 (15.8) years and 54.3% were male (Table 1). The mean (SD) age of onset of epilepsy was 39.2 (15.8) years. Patients had experienced a mean (SD) of 8.6 (21.7) seizures during the previous 3 months before DB baseline. Patient demographic and baseline characteristics were generally well balanced between treatment groups, including relevant medical conditions, such as CCV diseases. Overall, 10/96 (10.4%) patients in the ESL/ESL treatment group and 11/88 (12.5%) patients in the CBZ-CR/ESL treatment group were treated concomitantly with statins.

#### Changes in plasma lipid parameters

Mean (SD) levels of lipid parameters (total cholesterol, LDL cholesterol, HDL cholesterol and triglycerides) at DB baseline, OLE baseline and Table 1. Demographic and baseline characteristics. Adapted from Trinka et al.<sup>27</sup> with permission from Wiley.

Characteristic	ESL/ESL N=96	CBZ-CR/ESL N=88	Total <i>N</i> = 184	
Demographic characteristics <sup>a</sup>				
Sex – male, <i>n</i> (%)	55 (57.3)	45 (51.1)	100 (54.3)	
Age, mean (SD) years	42.5 (15.8)	41.6 (15.8)	42.1 (15.8)	
Ethnicity – Caucasian, <i>n</i> (%)	90 (93.8)	81 (92.0)	171 (92.9)	
Body mass index, mean (SD) kg/m <sup>2</sup>	25.7 (4.4)	25.9 (4.6)	25.8 (4.5)	
Epilepsy-related characteristics <sup>b</sup> Age at onset of epilepsy, mean (SD) years Time since last seizure, mean (SD) days Number of seizures during previous 3 months Total seizures Focal aware seizures Focal aware seizures Focal impaired awareness seizures Focal to bilateral tonic-clonic seizures Family history of epilepsy – yes, n (%) Concomitant use of statins, n (%)	39.5 (15.8) 19.3 (20.7) , mean (SD) 6.7 (12.8) 2.8 (6.7) 2.9 (11.8) 1.0 (0.9) 7 (7.3) 10 (10.4)	38.9 (15.8) 20.6 (22.9) 10.7 (28.3) 3.9 (8.8) 5.9 (27.9) 0.9 (1.1) 4 (4.5) 11 (12.5)	39.2 (15.8) 19.9 (21.7) 8.6 (21.7) 3.3 (7.8) 4.3 (21.1) 1.0 (1.0) 11 (6.0) 21 (11.4)	
Relevant medical history <sup>c</sup> Cardiac disorders, <i>n</i> (%) Angina pectoris, <i>n</i> (%) Cardiac failure chronic, <i>n</i> (%) Endocrine disorders, <i>n</i> (%) Hyperthyroidism, <i>n</i> (%) Metabolism and nutrition disorders, <i>n</i> (%) Dyslipidemia, <i>n</i> (%) Hypercholesterolemia, <i>n</i> (%) Hyperlipidemia, <i>n</i> (%) Obesity, <i>n</i> (%) Overweight, <i>n</i> (%) Nervous system disorders, <i>n</i> (%) Cerebrovascular accident, <i>n</i> (%) Surgical and medical procedures, <i>n</i> (%)	3 (3.1) 1 (1.0) 2 (2.1) 2 (2.1) 2 (2.1) 18 (18.8) 4 (4.2) 3 (3.1) 1 (1.0) 9 (9.4) 3 (3.1) 9 (9.4) 9 (9.4) 9 (9.4) 3 (3.1) 2 (2.1)	2 (2.3) 2 (2.3) 0 0 15 (17.0) 5 (5.7) 0 2 (2.3) 8 (9.1) 1 (1.1) 11 (12.5) 11 (12.5) 3 (3.4) 2 (2.4)	5 (2.7) 3 (1.6) 2 (1.1) 2 (1.1) 2 (1.1) 33 (17.9) 9 (4.9) 3 (1.6) 3 (1.6) 17 (9.2) 4 (2.2) 20 (10.9) 20 (10.9) 6 (3.3) ( (2.2)	
Vascular disorders, n (%) Arteriosclerosis, n (%) Hypertension, n (%)	23 (24.0) 2 (2.1) 23 (24.0)	20 (22.7) 0 20 (22.7)	43 (23.4) 2 (1.1) 43 (23.4)	

<sup>a</sup>At baseline of OLE study.

<sup>b</sup>At baseline of DB trial.

°Previous and concomitant medical conditions with  $\geq 2\%$  frequency in any group.

CBZ-CR, controlled-release carbamazepine; DB, double-blind; ESL, eslicarbazepine acetate; OLE, open-label extension; SD, standard deviation.

OLE EOS are presented in Table 2, and the proportions of patients with high/low levels of the lipid parameters at these timepoints are presented in Table 3 and Figures 3 and 4. Mean (SD) levels of lipid parameters in the subgroups of patients with high/low levels at DB baseline, OLE baseline

Parameter	ESL/ESL ( <i>N</i> =96 at DB baseline)	CBZ-CR/ESL ( <i>N</i> = 88 at DB baseline)	
Total cholesterol, mmol/L			
DB baseline, mean (SD)	5.0 (0.9)	5.1 (1.2)	
OLE baseline, mean (SD)	5.3 (1.0)	5.6 (1.3)	
OLE EOS, mean (SD)	5.1 (1.0)	5.2 (1.1)	
LDL cholesterol, mmol/L			
DB baseline, mean (SD)	3.1 (0.9)	3.2 (1.0)	
OLE baseline, mean (SD)	3.2 (0.9)	3.5 (1.2)	
OLE EOS, mean (SD)	3.2 (0.9)	3.3 (1.0)	
HDL cholesterol, mmol/L			
DB baseline, mean (SD)	1.4 (0.3)	1.4 (0.4)	
OLE baseline, mean (SD)	1.6 (0.5)	1.6 (0.5)	
OLE EOS, mean (SD)	1.6 (0.4)	1.6 (0.5)	
Triglycerides, mmol/L			
DB baseline, mean (SD)	1.4 (0.8)	1.3 (0.8)	
OLE baseline, mean (SD)	1.3 (0.8)	1.4 (0.9)	
OLE EOS, mean (SD)	1.3 (0.7)	1.2 (0.7)	

**Table 2.** Mean (SD) levels of total cholesterol, LDL cholesterol, HDL cholesterol and triglycerides at DB baseline, OLE baseline and OLE EOS (total monotherapy population).

CBZ-CR, controlled-release carbamazepine; DB, double-blind; EOS, end of study; ESL, eslicarbazepine acetate; HDL, high-density lipoprotein; LDL, low-density lipoprotein; OLE, open-label extension; SD, standard deviation.

and OLE EOS are presented in Supplemental Table S1. At DB baseline, there was no significant difference between treatment groups in the proportion of patients with high levels of total cholesterol [Figure 3(a) and Table 3]. At the end of the DB trial (OLE baseline), the proportion of patients with high levels of total cholesterol who had been treated with CBZ-CR had increased significantly from 15.9% to 31.0% (change from DB baseline to OLE baseline, 14.9%; p < 0.001), while in patients treated with ESL the proportion of patients with high levels of total cholesterol had increased non-significantly from 11.5% to 17.4% (change from DB baseline to OLE baseline, 6.5%; *p*=0.109) [Figure 3(a) and Table 3]. This resulted in a significant between-group difference (ESL-CBZ-CR) of -13.6% (p=0.037), demonstrating that a lower proportion of patients treated with ESL had high levels of total cholesterol at

OLE baseline, in comparison with those treated with CBZ-CR. In patients treated with CBZ-CR during the DB trial, the proportion of patients with high total cholesterol levels decreased significantly by OLE EOS after switching from CBZ-CR to ESL (change from OLE baseline to OLE EOS, -15.3%; *p*=0.008) [Figure 4(a) and Table 3]. By contrast, the proportion of patients with high total cholesterol levels remained relatively consistent in patients in the ESL/ESL treatment group during the DB trial and OLE study, without significant changes [Figure 4(a) and Table 3]. The cumulative risk of patients developing high levels of total cholesterol during the DB trial (having not previously had high levels) was 12.2% in those treated with ESL versus 19.2% in those treated with CBZ-CR [risk difference for ESL versus CBZ-CR, -7.0%; relative risk (95% CI), 0.64 (0.30-1.34); p=0.235; Table 4(a)].

#### THERAPEUTIC ADVANCES in Neurological Disorders

 Table 3. Proportions of patients with high/low lipid parameters (total monotherapy population).

Parameter	ESL/ESL ( <i>N</i> = 96 at DB baseline)	CBZ-CR/ESL ( <i>N</i> =88 at DB baseline)	ESL/ESL-CBZ-CR/ESL difference, % (p value)ª
Total cholesterol			
Patients with high levels <sup>b</sup> DB baseline, % ( $n/N$ )	11.5 (11/96)	15.9 (14/88)	-4.5 ( <i>p</i> =0.398)
Patients with high levels <sup>b</sup> OLE baseline, % ( <i>n/N</i> )	17.4 (16/92)	31.0 (27/87)	-13.6 ( <i>p</i> =0.037)
Patients with high levels <sup>b</sup> OLE EOS, % ( <i>n/N</i> )	11.8 (9/76)	17.8 (13/73)	-6.0 ( <i>p</i> =0.360)
Change from DB baseline to OLE baseline, % ( $p$ value) $^{ m c}$	6.5 ( <i>p</i> =0.109)	14.9 ( <i>p</i> < 0.001)	
Change from OLE baseline to OLE EOS, % ( $p$ value) $^{ m c}$	-2.7 ( <i>p</i> =0.527)	-15.3 ( <i>p</i> = 0.008)	
LDL cholesterol			
Patients with high levels <sup>d</sup> DB baseline, % ( $n/N$ )	14.6 (14/96)	14.8 (13/88)	-0.2 ( <i>p</i> = 1.000)
Patients with high levels <sup>d</sup> OLE baseline, % ( <i>n/N</i> )	14.1 (13/92)	26.4 (23/87)	-12.3 ( <i>p</i> =0.061)
Patients with high levels <sup>d</sup> OLE EOS, % ( <i>n/N</i> )	14.5 (11/76)	15.1 (11/73)	-0.6 ( <i>p</i> =1.000)
Change from DB baseline to OLE baseline, % ( $p$ value) $^{ m c}$	0 ( <i>p</i> = 1.000)	11.5 ( <i>p</i> = 0.012)	
Change from OLE baseline to OLE EOS, % ( $p$ value) $^{ m c}$	1.4 ( <i>p</i> =0.706)	-11.1 ( <i>p</i> = 0.021)	
HDL cholesterol			
Patients with low levels $^{\circ}$ DB baseline, $\%$ ( <i>n/N</i> )	6.3 (6/96)	18.2 (16/88)	-11.9 ( <i>p</i> =0.021)
Patients with low levels $^{\circ}$ OLE baseline, % ( <i>n/N</i> )	4.3 (4/92)	10.3 (9/87)	-6.0 ( <i>p</i> =0.154)
Patients with low levels $^{\circ}$ OLE EOS, % ( <i>n/N</i> )	3.9 (3/76)	12.3 (9/73)	-8.4 ( <i>p</i> =0.075)
Change from DB baseline to OLE baseline, % (p value) <sup>c</sup>	-1.1 ( <i>p</i> =0.655)	-8.1 ( <i>p</i> =0.052)	
Change from OLE baseline to OLE EOS, % (p value) <sup>c</sup>	-1.4 ( <i>p</i> =0.564)	2.8 ( <i>p</i> =0.414)	
Triglycerides			
Patients with high levels <sup>f</sup> DB baseline, $\%$ ( <i>n</i> / <i>N</i> )	25.0 (24/96)	18.2 (16/88)	6.8 ( <i>p</i> = 0.287)
Patients with high levels OLE baseline, $\%$ ( <i>n</i> / <i>N</i> )	22.8 (21/92)	19.5 (17/87)	3.3 ( <i>p</i> = 0.715)
Patients with high levels <sup><math>f</math></sup> OLE EOS, % ( $n/N$ )	22.4 (17/76)	15.1 (11/73)	7.3 ( <i>p</i> = 0.298)
Change from DB baseline to OLE baseline, % ( <i>p</i> value) <sup>c</sup>	-1.1 ( <i>p</i> =0.796)	1.2 ( <i>p</i> = 0.808)	
Change from OLE baseline to OLE EOS, % ( <i>p</i> value) <sup>c</sup>	-1.4 ( <i>p</i> =0.796)	-1.4 ( <i>p</i> =0.796)	

<sup>a</sup>Fisher's test was conducted to assess if the proportion of patients with high/low levels of corresponding lipid parameter differed significantly between treatment arms. *p* values use standard alpha level (0.05).

<sup>b</sup>High level of total cholesterol defined as > 6.22 mmol/L.

<sup>c</sup>McNemar test was conducted to assess if the treatment used between visits impacted the proportion of patients (using only completer subjects with data in both visits) with high/low levels of corresponding lipid parameter. *p* values use standard alpha level (0.05).

<sup>d</sup>High level of LDL cholesterol defined as > 4.14 mmol/L.

 $^{\rm e}$ Low level of HDL cholesterol defined as < 1.04 mmol/L.

<sup>f</sup>High level of triglycerides defined as > 1.70 mmol/L.

Results in bold represent statistically significant change.

CBZ-CR, controlled-release carbamazepine; DB, double-blind; EOS, end of study; ESL, eslicarbazepine acetate; HDL, high-density lipoprotein; LDL, low-density lipoprotein; OLE, open-label extension.



**Figure 3.** Between-group comparisons of the proportions of patients at DB baseline, OLE baseline and OLE EOS with (a) high levels of total cholesterol (>6.22 mmol/L), (b) high levels of LDL cholesterol (>4.14 mmol/L), (c) low levels of HDL cholesterol (<1.04 mmol/L), and (d) high levels of triglycerides (>1.70 mmol/L). (a) Patients who received ESL the DB trial and OLE study. (b) Patients who received CBZ-CR in the DB trial and ESL in the OLE study.

CBZ-CR, controlled-release carbamazepine; DB, double-blind; EOS, end of study; ESL, eslicarbazepine acetate; HDL, high-density lipoprotein; LDL, low-density lipoprotein; NS, not significant; OLE, open-label extension.

Similarly, at DB baseline there was no significant difference between treatment groups in the proportion of patients with high LDL cholesterol levels [Figure 3(b) and Table 3]. By the end of the DB trial (OLE baseline), the proportion of patients with high levels of LDL cholesterol was higher in patients treated with CBZ-CR during the DB trial, compared with those treated with ESL, although the difference was not statistically significant [between-group difference (ESL-CBZ-CR), -12.3%; p=0.061]. The proportion of patients with high LDL cholesterol was similar between treatment groups at the other two visits (DB baseline and OLE EOS), showing that the increased proportion of patients with high LDL after using CBZ-CR had decreased to the baseline proportion after switching to ESL [Figure 3(b) and Table 3]. During the DB trial, the proportion of patients with high LDL cholesterol increased significantly in patients treated with CBZ-CR (change from DB baseline to OLE baseline, 11.5%; p = 0.012), and additionally, after switching from CBZ-CR to ESL, decreased significantly by OLE EOS (change from OLE baseline to OLE EOS, -11.1%; p=0.021) [Figure 4(b) and Table 3]. In contrast, the proportion of patients with high LDL cholesterol remained stable in patients in the ESL/ESL treatment group during the DB trial and OLE study, without significant changes [Figure 4(b) and Table 3]. The cumulative risk of patients developing high levels of

#### THERAPEUTIC ADVANCES in Neurological Disorders



**Figure 4.** Within-group comparisons of the proportions of patients at DB baseline, OLE baseline and OLE EOS with (a) high levels of total cholesterol (>6.22 mmol/L), (b) high levels of LDL cholesterol (>4.14 mmol/L), (c) low levels of HDL cholesterol (<1.04 mmol/L), and (d) high levels of triglycerides (>1.70 mmol/L). (a) Patients who received ESL during the DB trial and OLE study. (b) Patients who received CBZ-CR in the DB trial and ESL in the OLE study.

CBZ-CR, controlled-release carbamazepine; DB, double-blind; EOS, end of study; ESL, eslicarbazepine acetate; HDL, high-density lipoprotein; LDL, low-density lipoprotein; NS, not significant; OLE, open-label extension.

LDL cholesterol during the DB trial (having not previously had high levels) was 7.6% in those treated with ESL *versus* 17.6% in those treated with CBZ-CR [risk difference for ESL *versus* CBZ-CR, -10.0%; relative risk (95% CI), 0.43 (0.17–1.08); p = 0.072; Table 4(a)].

At DB baseline, the proportion of patients with low HDL cholesterol was significantly higher in patients randomized to CBZ-CR than in those randomized to ESL [between-group difference (ESL-CBZ-CR), -11.9; p=0.021] [Figure 3(c) and Table 3]. In patients treated with ESL during the DB trial, the proportion with low HDL cholesterol remained relatively stable during both the DB trial and OLE study. In contrast, in patients treated with CBZ-CR during the DB trial, there was a non-significant decrease in the proportion with low HDL cholesterol during the DB trial, which remained relatively stable after switching to ESL during the OLE study [Figures 3(c), 4(c) and Table 3]. However, between-group differences observed during the DB trial and OLE study are inconclusive, due to the differences at DB baseline originally presented.

There were no significant changes in the proportion of patients with high triglyceride levels within treatment arms during the course of the DB trial and OLE study, and no significant **Table 4.** Risk of patients with normal levels of total cholesterol and LDL cholesterol at DB baseline developing high levels of these lipid parameters following treatment with ESL and CBZ-CR in the DB trial: (A) total monotherapy population and (B) subgroup of patients not treated concomitantly with statins.

Parameter	ESL/ESL (N=96 at DB baseline)	CBZ-CR/ESL ( <i>N</i> =88 at DB baseline)	ESL/ESL-CBZ-CR/ ESL difference
(A) Total monotherapy population			
Total cholesterol <sup>a</sup>			
Cumulative risk (cumulative incidence), %	12.2	19.2	-
Risk difference, %	-	-	-7.0
Relative risk (95% CI); <i>p</i> value	-	-	0.64 (0.30, 1.34); <i>p</i> =0.235
LDL cholesterol <sup>b</sup>			
Cumulative risk (cumulative incidence), %	7.6	17.6	-
Risk difference, %	-	-	-10.0
Relative risk (95% CI); <i>p</i> value	-	-	0.43 (0.17, 1.08); p=0.072
(B) Subgroup of patients not treated concomitantly with statins			
Total cholesterol <sup>a</sup>			
Cumulative risk (cumulative incidence), %	12.5	19.4	-
Risk difference, %	-	-	-6.9
Relative risk (95% CI); <i>p</i> value	-	-	0.64 (0.30, 1.41); p=0.271
LDL cholesterol <sup>b</sup>			
Cumulative risk (cumulative incidence), %	8.6	19.1	-
Risk difference, %	-	-	-10.6
Relative risk (95% CI); <i>p</i> value	-	-	0.45 (0.18, 1.11); <i>p</i> =0.083
<sup>a</sup> High level of total cholesterol defined a	as $> 6.22 \text{ mmol/L}$		

CBZ-CR, controlled-release carbamazepine; DB, double-blind; ESL, eslicarbazepine acetate; HDL, high-density

lipoprotein; LDL, low-density lipoprotein; OLE, open-label extension.

between-group differences [Figures 3(d), 4(d) and Table 3].

Analyses of the subset of patients who were not being treated concomitantly with statins were generally consistent with those observed for the total monotherapy population (Table 5). During the DB trial, the proportion of patients with high total cholesterol and high LDL cholesterol increased significantly in those treated with CBZ-CR but not significantly in those treated with ESL. Unlike in the total monotherapy population, the proportion of patients with high levels of total cholesterol at the end of the DB trial (i.e. at OLE baseline) did not

differ significantly between treatment groups [between-group difference (ESL-CBZ-CR), -9.3%; p = 0.187]. During the OLE study, the proportions of patients with high total cholesterol and high LDL cholesterol significantly decreased in those switching from CBZ-CR to ESL, while remaining stable in those who received ESL throughout the DB trial and OLE study. The cumulative risk of patients developing high levels of total cholesterol during the DB trial (having not previously had high levels) was 12.5% in those treated with ESL versus 19.4% in those treated with CBZ-CR [risk difference for ESL versus CBZ-CR, -6.9%; relative risk (95% CI), 0.64 (0.30–1.41); p = 0.271; Table 4(b)]. The cumulative risk of patients developing high levels of LDL cholesterol during the DB trial (having not previously had high levels) was 8.6% in those treated with ESL versus 19.1% in those treated with CBZ-CR [risk difference for ESL versus CBZ-CR, -10.6%; relative risk (95% CI), 0.45 (0.18-1.11); p = 0.083; Table 4(b)].

As in the total monotherapy population, the proportion of patients with low HDL cholesterol was significantly higher in patients randomized to CBZ-CR than in those randomized to ESL (Table 5). Within-group trends were consistent with those observed in the total monotherapy population. At OLE EOS, the proportion of patients with low levels of HDL cholesterol was significantly lower in patients randomized to ESL in the DB trial, in comparison with those who were randomized to CBZ-CR [between-group difference (ESL-CBZ-CR), -9.3%; p=0.050]. However, as in the total monotherapy population, between-group differences were inconclusive due to the differences at DB baseline.

As in the total monotherapy population, there were no significant within-group changes or between-group differences in the proportion of patients with high triglyceride levels (Table 5).

#### Discussion

In this *post hoc* analysis of a phase III DB trial and OLE study, the proportions of patients with elevated levels of total cholesterol and LDL

cholesterol increased significantly in those treated with CBZ-CR monotherapy in the DB trial and decreased significantly after switching from CBZ-CR to ESL monotherapy during 2 years of treatment in the OLE study. There were minimal changes in the proportions of patients with elevated levels of total cholesterol and LDL cholesterol in those treated with ESL monotherapy throughout the DB trial and OLE study (up to 30 months). The proportion of patients with high levels of total cholesterol at the end of the DB trial was significantly lower for those treated with ESL than for those treated with CBZ-CR, and there was a trend toward a lower proportion of patients with high levels of LDL cholesterol in those treated with ESL, in comparison with CBZ-CR. At the end of the OLE study, the between-group differences for both these parameters were not significant. Additional analyses to further understand changes in total cholesterol and LDL cholesterol were conducted by estimating the risk of developing high levels of total cholesterol and LDL cholesterol separately during the DB trial, and assessing the relative risk reduction for ESL versus CBZ-CR. This involved evaluating those patients who did not have high levels at DB baseline and assessing whether their levels of total and LDL cholesterol increased during the DB trial. Although the findings may indicate trends toward a lower risk of developing high total and LDL cholesterol when treated with ESL monotherapy, in comparison with CBZ-CR monotherapy, the limitation is that these analyses were only exploratory and no multiplicity correction of p values was conducted, and they should therefore be interpreted with caution and only considered within the context of this post hoc evaluation. Since the proportion of patients with low levels of HDL cholesterol differed significantly between treatment groups at baseline of the DB trial, it was not possible to draw meaningful conclusions from the findings for this lipid parameter. No notable withingroup changes or between-group differences were observed for the proportions of patients with elevated triglyceride levels.

Statins are the most commonly used lipid-lowering agents and there is concern that the concomitant use of enzyme-inducing ASMs may affect the lipid-lowering efficacy of statins by inducing their metabolism.<sup>6,7</sup> In the current study, the proportion of patients treated concomitantly with statins **Table 5.** Subgroup analyses of proportions of patients with high/low lipid parameters in patients not treated concomitantly with statins.

Pa	rameter	ESL/ESL ( <i>N</i> =86 at DB baseline)	CBZ-CR/ESL ( <i>N</i> =77 at DB baseline)	ESL/ESL-CBZ-CR/ESL difference, % (p value)ª
То	tal cholesterol			
	Patients with high levels <sup>b</sup> DB baseline, % ( $n/N$ )	12.8 (11/86)	11.7 (9/77)	1.1 ( <i>p</i> = 1.000)
	Patients with high levels <sup>b</sup> OLE baseline, % ( $n/N$ )	18.3 (15/82)	27.6 (21/76)	-9.3 ( <i>p</i> =0.187)
	Patients with high levels <sup>b</sup> OLE EOS, % ( $n/N$ )	12.9 (9/70)	16.7 (11/66)	-3.8 ( <i>p</i> =0.630)
	Change from DB baseline to OLE baseline, % (p value) <sup>c</sup>	6.1 ( <i>p</i> =0.166)	15.8 ( <i>p</i> = 0.001)	
	Change from OLE baseline to OLE EOS, % (p value) <sup>c</sup>	-2.9 ( <i>p</i> =0.527)	-13.9 ( <i>p</i> =0.013)	
LDL cholesterol				
	Patients with high levels <sup>d</sup> DB baseline, % $(n/N)$	15.1 (13/86)	10.4 (8/77)	4.7 ( <i>p</i> =0.484)
	Patients with high levels <sup>d</sup> OLE baseline, % ( $n/N$ )	14.6 (12/82)	23.7 (18/76)	-9.1 ( <i>p</i> =0.161)
	Patients with high levels <sup>d</sup> OLE EOS, % ( $n/N$ )	15.7 (11/70)	13.6 (9/66)	2.1 ( <i>p</i> =0.811)
	Change from DB baseline to OLE baseline, % (p value) <sup>c</sup>	0 ( <i>p</i> = 1.000)	13.2 ( <i>p</i> = 0.012)	
	Change from OLE baseline to OLE EOS, % (p value) <sup>c</sup>	1.5 ( <i>p</i> =0.706)	-10.8 ( <i>p</i> =0.020)	
Н	DL cholesterol			
	Patients with low levels <sup>e</sup> DB baseline, % ( $n/N$ )	4.7 (4/86)	15.6 (12/77)	-10.9 ( <i>p</i> = 0.032)
	Patients with low levels <sup>e</sup> OLE baseline, % ( <i>n/N</i> )	3.7 (3/82)	10.5 (8/76)	-6.9 ( <i>p</i> =0.120)
	Patients with low levels <sup>e</sup> OLE EOS, % ( <i>n/N</i> )	2.9 (2/70)	12.1 (8/66)	-9.3 ( <i>p</i> =0.050)
	Change from DB baseline to OLE baseline, % (p value) <sup>c</sup>	0 ( <i>p</i> = 1.000)	-5.3 ( <i>p</i> =0.206)	
	Change from OLE baseline to OLE EOS, % (p value) <sup>c</sup>	2.9 ( <i>p</i> =0.157)	3.1 ( <i>p</i> =0.414)	
Triglycerides				
	Patients with high levels <sup>f</sup> DB baseline, % ( <i>n/N</i> )	20.9 (18/86)	15.6 (12/77)	5.4 ( <i>p</i> =0.423)
	Patients with high levels <sup>f</sup> OLE baseline, % ( <i>n/N</i> )	19.5 (16/82)	17.1 (13/76)	2.4 ( <i>p</i> =0.837)
	Patients with high levels <sup>f</sup> OLE EOS, % ( $n/N$ )	20.0 (14/70)	15.2 (10/66)	4.9 ( <i>p</i> =0.506)
	Change from DB baseline to OLE baseline, % (p value) <sup>c</sup>	0 ( <i>p</i> = 1.000)	1.3 ( <i>p</i> = 0.796)	
	Change from OLE baseline to OLE EOS, % ( <i>p</i> value) <sup>c</sup>	-1.5 ( <i>p</i> =0.782)	-1.5 ( <i>p</i> =0.782)	

<sup>a</sup>Fisher's test was conducted to assess if the proportion of patients with high/low levels of corresponding lipid parameter differed significantly between treatment arms. *p* values use standard alpha level (0.05).

<sup>b</sup>High level of total cholesterol defined as >6.22 mmol/L.

<sup>c</sup>McNemar test was conducted to assess if the treatment used between visits impacted the proportion of patients (using only completer subjects with data in both visits) with high/low levels of corresponding lipid parameter. *p* values use standard alpha level (0.05).

 $^{\rm d}{\rm High}$  level of LDL cholesterol defined as  ${>}4.14\,\rm{mmol/L}.$ 

 $^{\rm e}\text{Low}$  level of HDL cholesterol defined as  ${<}1.04\,\text{mmol/L}.$ 

 $^{\rm f}{\rm High}$  level of triglycerides defined as  ${>}1.70\,{\rm mmol/L}.$ 

Results in bold represent statistically significant change.

CBZ-CR, controlled-release carbamazepine; DB, double-blind; EOS, end of study; ESL, eslicarbazepine acetate; HDL, high-density lipoprotein; LDL, low-density lipoprotein; OLE, open-label extension.

was too small to be analyzed separately [ESL/ ESL subgroup, 10/96 (10.4%); CBZ-CR/ESL subgroup, 11/88 (12.5%)] and so it was not possible to assess the potential impact of ESL *versus* CBZ-CR on the lipid-lowering efficacy of statins. However, the findings from the subset of patients who were not being concomitantly treated with statins were generally consistent with those for the total monotherapy population.

Although both ESL and CBZ belong to the dibenzazepine family of sodium-channel-blocking ASMs, sharing a dibenzazepine nucleus bearing the 5-carboxamide substituent,<sup>29</sup> they differ pharmacologically in terms of metabolism, pharmacokinetics and pharmacodynamics.<sup>19,30</sup> Pharmacodynamically, S-licarbazepine (the primary active metabolite of ESL, accounting for 94% of plasma drug exposure following oral administration) is thought to act primarily by enhancing the slow inactivation of voltage-gated sodium channels, whereas CBZ is thought to alter the fast inactivation of these channels.<sup>19,31</sup> Moreover, S-licarbazepine appears to have greater selectivity than CBZ for inhibiting rapidly firing active 'epileptic' neurons.<sup>19,32</sup> CBZ's tolerability is compromised by being metabolized to carbamazepine-10,11-epoxide, which, in addition to its antiseizure effect, is known to cause adverse effects, including dizziness and somnolence.19,30 Furthermore, CBZ is a potent inducer of the cytochrome P450 (CYP450) enzyme system, which not only negatively affects the duration and action of many drugs but can also impact enzymes involved in endogenous metabolic pathways.<sup>10,12,19</sup> By contrast, S-licarbazepine is a less potent enzyme inducer than CBZ,<sup>14</sup> being a weak inducer of CYP450 3A4 and uridine 5'-diphospho-glucuronosyl transferases.<sup>19,23,25</sup> The impact of CBZ on endogenous metabolic pathways can lead to CCV diseases and other comorbidities, including sexual dysfunction and osteoporosis.<sup>10,12</sup> Importantly, the CYP450 enzyme system is extensively involved in the synthesis and metabolism of cholesterol<sup>33</sup> and it is thought that potent CYP450 inducers, such as CBZ, increase cholesterol production by reducing levels of oxysterol intermediates, which are the main endogenous feedback inhibitors of HMG-CoA reductase (the rate-controlling enzyme in cholesterol synthesis).<sup>10,34</sup> CBZ additionally causes alterations in thyroid hormone levels, which may also cause changes in serum lipid levels.35,36 Evidence for the

impact of CBZ on lipid levels is substantial<sup>7,10</sup>; for example, in an analysis of data from a 12-month, phase III, international, randomized, DB trial of CBZ versus lacosamide for the initial treatment of focal epilepsy, CBZ-treated patients experienced increased levels of total cholesterol (+21.1 mg/ dL), LDL cholesterol (+12.6 mg/dL), non-HDL cholesterol (+12.5 mg/dL) and HDL cholesterol  $(+8.5 \, \text{mg/dL}),$ whereas lacosamide-treated patients did not experience elevations in these parameters.<sup>37</sup> Moreover, the proportion of patients with elevated total cholesterol levels increased from 30.8% at baseline to 49.6% at 12 months in patients treated with CBZ but remained stable in those treated with lacosamide (37.0% at baseline; 34.8% at 12 months).<sup>37</sup> In a study that assessed the impact of discontinuation of CBZ due to concerns over the long-term consequences of enzyme induction, patients who discontinued CBZ were shown to experience significant reductions in total cholesterol, LDL cholesterol and HDL cholesterol, and to additionally experience a significant increase in free testosterone and a significant decrease in sex hormone-binding globulin levels, in comparison with those who continued to receive CBZ.38 Another study demonstrated that CBZ significantly affected the ability of statins to lower total cholesterol, leading the authors to conclude that it is a poor choice for patients with hyperlipidemia or cardiovascular disease.7 Such findings have led to recommendations regarding the consideration of switching patients treated with enzyme-inducing ASMs to non-enzyme-inducing ASMs, 10, 26, 38, 39 and to speculation regarding whether or not CBZ should be considered as first-line monotherapy in patients with newly diagnosed epilepsy.10

The findings of the current study are consistent with previous reports.<sup>6,19–21,24</sup> A single-center, observational, retrospective cohort study assessed the effects of ESL on lipid metabolism in 108 patients (without having newly diagnosed focal seizures, and including patients with add-on ASMs) over a median duration of 23.1 months (range 3–41 months), and, in the overall population, mean total cholesterol levels decreased significantly during follow-up, changing from previous increased levels (>200 mg/dL) to normal levels (<200 mg/dL) during treatment.<sup>20</sup> The proportion of patients with hypercholesterolemia also decreased during ESL treatment, although this change was not statistically significant.<sup>20</sup> Over half of the population had switched to ESL from prior treatment with CBZ or oxcarbazepine, and there was a marked decrease in mean total cholesterol and a significant decrease in both mean LDL cholesterol and median triglyceride levels in these patients.<sup>20</sup> Furthermore, after transitioning to ESL, there were significant reductions in the proportions of patients with hypercholesterolemia and hypertriglyceridemia.<sup>20</sup> In another retrospective cohort study, lipid levels were assessed in 36 adult patients before and after receiving treatment with adjunctive ESL over a mean follow-up duration of 11 months (range 6–18 months).<sup>21</sup> After at least 6 months of ESL treatment, there were significant reductions in the mean levels of total cholesterol and LDL cholesterol, and a significant increase in the mean level of HDL cholesterol.<sup>21</sup> As in the current study, no statistically significant change in triglyceride levels was observed.<sup>21</sup> In a post hoc analysis of pooled data from three phase III, multicenter, randomized, DB, placebo-controlled trials of adjunctive ESL therapy (400, 800, or 1200 mg OD) in patients with refractory focal seizures, changes from baseline in serum lipid levels were analyzed according to the use of enzyme-inducing ASMs and/or statins during the baseline period.<sup>6</sup> The most frequently used enzyme-inducing ASM in patients treated with ESL was CBZ (51%).6 Changes from baseline were significantly greater for ESL 1200 mg/day versus placebo for total cholesterol and HDL cholesterol, but the effect sizes versus placebo were small.<sup>6</sup> There was also a small but significant difference in the change from baseline in LDL cholesterol for ESL 400 mg/day versus placebo.6 ESL had no consistent effect on lipid levels in patients taking a concomitant enzyme-inducing ASM, and there were no clinically relevant changes in lipid levels following ESL treatment in patients who took statins during baseline.<sup>6</sup> In a similar post hoc analysis of pooled data from two phase III conversion to ESL monotherapy studies, lipid parameters were assessed at baseline, Week 18, and Month 12, and analyzed in terms of the use of enzyme-inducing ASMs at baseline.<sup>24</sup> The most frequently used enzyme-inducing ASMs were CBZ (77%) and phenytoin (24%).24 Patients converting from enzyme-inducing ASMs had higher levels of total cholesterol, LDL cholesterol, HDL cholesterol and triglycerides at baseline, compared with those converting from non-enzymeinducing ASMs (significantly higher for total cholesterol and HDL cholesterol).24 After converting to ESL monotherapy, plasma levels of total

cholesterol, LDL cholesterol, HDL cholesterol and triglycerides decreased in patients converting from enzyme-inducing ASMs and increased in those converting from non-enzyme-inducing ASMs, reaching very similar values that were intermediate between the baseline values for the subgroups of patients converting from enzyme-inducing and non-enzyme-inducing ASMs.<sup>24</sup> The authors concluded that ESL had only a modest and clinically insignificant impact on plasma lipid levels.<sup>24</sup> Other studies, conducted in the clinical practice setting, have demonstrated that ESL monotherapy can be effective in patients who have switched from CBZ due to inadequate efficacy,<sup>40,41</sup> consistent with evidence demonstrating that ESL and CBZ have different mechanisms of action.<sup>19,31,32</sup> Since no ASM has been shown to have protective CCV effects, in terms of improving lipid profiles in previously untreated patients, the relative impact of an ASM on CCV risk factors is an important consideration when choosing initial treatment, or when considering switching a patient from one ASM to another. Published guidance has provided recommendations for clinical scenarios in which it may be appropriate to switch patients from CBZ to ESL (such as patients experiencing or at risk of hypercholesterolemia and those with severe osteopenia-osteoporosis), along with pragmatic advice on how to do so.26,39

To our knowledge, this is the first prospective investigation of the impact of ESL on plasma lipid levels in patients receiving ESL as monotherapy for newly diagnosed epilepsy (i.e. without prior exposure to ASMs), since previous studies were conducted either in the adjunctive or conversion to monotherapy settings and/or were retrospective in design. A further strength is that we were able to analyze patients' plasma lipid levels before starting ASM treatment, and after switching from CBZ monotherapy to ESL monotherapy. The nature of this post hoc study was exploratory. The sample size was small, and the original DB trial and OLE study were not designed or powered to assess the impact of treatment on plasma lipid levels. Therefore, the p values cited should not be considered confirmatory and the findings from the analyses should be interpreted with caution. Nevertheless, the results are consistent with those of previous studies.6,19-21,24 Since the sample size was small, it was not possible to conduct meaningful analyses of mean changes in lipid parameters, due to high variability in the data, but this was addressed by assessing the proportions of patients with high/low plasma levels of lipid parameters to provide a clinically meaningful understanding of changes in these parameters over time. A further limitation of the study was that information on alcohol use and smoking status, both of which can have an impact on lipid parameters, was not available, and it was therefore not possible to confirm whether there were between-group differences that could potentially have affected the results. However, relevant medical conditions, including CCV diseases, were well balanced between the treatment groups.

In summary, the findings from this study are consistent with current evidence regarding the impact of CBZ on plasma lipid levels, supporting recommendations for considering switching patients from CBZ to ESL to reduce the risk of hypercholesterolemia.<sup>19,38,39</sup> The findings also suggest that ESL monotherapy may be a rational treatment option for patients with newly diagnosed focal epilepsy who have, or who are at risk of developing, hypercholesterolemia.

#### Declarations

#### Ethics approval and consent to participate

The study was conducted in accordance with the Declaration of Helsinki and International Council for Harmonization Good Clinical Practice guidelines and approved in all participating countries by the Regulatory Authorities and Central Ethics Committees before patient enrollment was commenced (the full list is reported in the Supplemental Information). All patients provided written informed consent before undergoing any study-related activities.

*Consent for publication* Not applicable.

#### Author contributions

**Eugen Trinka:** Conceptualization; Formal analysis; Writing – original draft; Writing – review & editing.

**Rodrigo Rocamora:** Writing – review & editing.

João Chaves: Writing – review & editing.

Mathias J. Koepp: Writing – review & editing.

Stephan Rüegg: Writing – review & editing.

Martin Holtkamp: Writing – review & editing.

**Joana Moreira:** Conceptualization; Data curation; Formal analysis; Funding acquisition; Investigation; Methodology; Project administration; Resources; Supervision; Validation; Visualization; Writing – original draft; Writing – review & editing.

**Miguel M. Fonseca:** Conceptualization; Data curation; Formal analysis; Funding acquisition; Investigation; Methodology; Project administration; Resources; Supervision; Validation; Visualization; Writing – original draft; Writing – review & editing.

**Guillermo Castilla-Fernández:** Conceptualization; Data curation; Formal analysis; Funding acquisition; Investigation; Methodology; Project administration; Resources; Supervision; Validation; Visualization; Writing – original draft; Writing – review & editing.

**Fábio Ikedo:** Conceptualization; Data curation; Formal analysis; Funding acquisition; Investigation; Methodology; Project administration; Resources; Supervision; Validation; Visualization; Writing – original draft; Writing – review & editing.

#### Acknowledgements

Third party submissions: not applicable. Medical writing assistance (editing, collating of author comments, formatting to journal style) was provided by John Scopes of mXm Medical Communications and funded by Bial – Portela & C<sup>a</sup>.

#### Funding

The authors disclosed receipt of the following financial support for the research, authorship, and/or publication of this article: This study was funded by Bial – Portela & C<sup>a</sup>.

#### Competing interests

ET reports personal fees from EVER Pharma, Marinus, Arvelle, Angelini, Argenx, Medtronic, Bial-Portela & C<sup>a</sup>, NewBridge, GL Pharma, GlaxoSmithKline, Boehringer Ingelheim, LivaNova, Eisai, UCB, Biogen, Sanofi, Jazz Pharmaceuticals, and Actavis. His institution received grants from Biogen, UCB Pharma, Eisai, Red Bull, Merck, Bayer, the European Union, FWF Osterreichischer Fond zur Wissenschaftsforderung, Bundesministerium für Wissenschaft

und Forschung. and Iubiläumsfond der Österreichischen Nationalbank. RR is a consultant for Eisai, Bial, UCB, GlaxoSmithKline and Shire, and receives grant and research support from Bial, Eisai, and UCB. JC has received speaker's honoraria and/or consultancy fees from Bial and Eisai and was granted with a Tecnifar Bursary. MJK reports personal fees as speaker or consultant from Arvelle, BIAL, Eisai, GW Pharmaceuticals, Novartis and UCB Pharma. He holds stocks for Prevep. MJK or his institution received grants from UCB Pharma, Eisai, Arvelle/Angelini, the European Union, Wellcome Trust and Medical Research Council. SR received unconditional research grants from UCB-pharma, honoraria from serving on the scientific advisory boards of Arvelle/Angelini, Bial, Eisai, GW/Jazz Pharmaceuticals, and UCB Pharma, and from serving as a consultant for Arvelle/ Angelini, Eisai, Jazz Pharmaceuticals, Pfizer, Novartis, Sandoz, and UCB Pharma. He does not hold any stocks of any pharmaceutical industries or manufacturers of medical devices. He received funding from UCB Pharma, and the Swiss National Science Foundation. MH received speaker's honoraria and/or consultancy fees from Angelini/Arvelle, Bial, Desitin, Eisai, GW Pharmaceuticals/Jazz, Neuraxpharm, UCB, and Zogenix with the last 3 years. JM, MMF and GCF are current employees of Bial - Portela & Ca, S.A. FI was an employee of Bial – Portela & C<sup>a</sup>, S.A. at the time this study was conducted but is currently employed at Neuroscience Clinical Development, Biopharmaceuticals R&D, AstraZeneca, Cambridge, UK.

#### Availability of data and materials

Not applicable.

#### **ORCID iD**

Eugen Trinka D https://orcid.org/0000-0002-5950-2692

#### Supplemental material

Supplemental material for this article is available online.

#### References

 European Medicines Agency. Zebinix®

 (eslicarbazepine acetate) Summary of Product Characteristics, https://www.ema.europa.eu/en/ documents/product-information/zebinix-eparproduct-information\_en.pdf (2022, accessed 29 August 2023).

- Food and Drug Administration. Aptiom® (eslicarbazepine acetate) Prescribing Information, https://www.accessdata.fda.gov/drugsatfda\_docs/ label/2019/022416s0111bl.pdf (2019, accessed 29 August 2023).
- Trinka E, Ben-Menachem E, Kowacs PA, et al. Efficacy and safety of eslicarbazepine acetate versus controlled-release carbamazepine monotherapy in newly diagnosed epilepsy: a phase III double-blind, randomized, parallel-group, multicenter study. *Epilepsia* 2018; 59: 479–491.
- 4. Hamed SA, Hamed EA, Hamdy R, *et al.* Vascular risk factors and oxidative stress as independent predictors of asymptomatic atherosclerosis in adult patients with epilepsy. *Epilepsy Res* 2007; 74: 183–192.
- 5. Katsiki N, Mikhailidis DP and Nair DR. The effects of antiepileptic drugs on vascular risk factors: a narrative review. *Seizure* 2014; 23: 677–684.
- 6. Mintzer S, Wechsler RT, Rogin JB, *et al.* Effects of adjunctive eslicarbazepine acetate on serum lipids in patients with partial-onset seizures: impact of concomitant statins and enzyme-inducing antiepileptic drugs. *Epilepsy Res* 2018; 141: 83–89.
- Mintzer S, Trinka E, Kraemer G, et al. Impact of carbamazepine, lamotrigine, and levetiracetam on vascular risk markers and lipidlowering agents in the elderly. *Epilepsia* 2018; 59: 1899–1907.
- 8. Terman SW, Aubert CE, Hill CE, *et al.* Cardiovascular disease risk, awareness, and treatment in people with epilepsy. *Epilepsy Behav* 2021; 117: 107878.
- Laxer KD, Trinka E, Hirsch LJ, et al. The consequences of refractory epilepsy and its treatment. *Epilepsy Behav* 2014; 37: 59–70.
- Mintzer S and Mattson RT. Should enzymeinducing antiepileptic drugs be considered firstline agents? *Epilepsia* 2009; 50(Suppl. 8): 42–50.
- Chuang YC, Chuang HY, Lin TK, et al. Effects of long-term antiepileptic drug monotherapy on vascular risk factors and atherosclerosis. *Epilepsia* 2012; 53: 120–128.
- Brodie MJ, Mintzer S, Pack AM, et al. Enzyme induction with antiepileptic drugs: cause for concern? *Epilepsia* 2013; 54: 11–27.
- Fox J, Ajinkya S and Lekoubou A. Enzymeinducing antiseizure medication utilization in patients with epilepsy and vascular risk factors. *Epilepsy Behav* 2020; 112: 107465.

- Josephson CB, Wiebe S, Delgado-Garcia G, et al. Association of enzyme-inducing antiseizure drug use with long-term cardiovascular disease. *JAMA Neurol* 2021; 78: 1367–1374.
- 15. Vossler DG. Stroke and ischemic heart disease with enzyme-inducing antiseizure medications: time to change prescribing habits. *Epilepsy Curr* 2022; 22: 120–122.
- Tseng WJ, Chang CW, Hwang JS, et al. Association of long-term antiseizure medication use and incident type 2 diabetes mellitus. *Neurology* 2023; 100: e2071–e2082.
- Larsson D, Baftiu A, Johannessen Landmark C, et al. Association between antiseizure drug monotherapy and mortality for patients with poststroke epilepsy. *JAMA Neurol* 2022; 79: 169–175.
- Hsu CY, Cheng CY, Lee JD, et al. Effects of long-term anti-seizure medication monotherapy on all-cause death in patients with post-stroke epilepsy: a nationwide population-based study in Taiwan. BMC Neurol 2021; 21: 226.
- Lawthom C, Peltola J, McMurray R, et al. Dibenzazepine agents in epilepsy: how does eslicarbazepine acetate differ? *Neurol Ther* 2018; 7: 195–206.
- Ley M, Principe A, Jiménez-Conde J, et al. Assessing long-term effects of eslicarbazepine acetate on lipid metabolism profile, sodium values and liver function tests. *Epilepsy Res* 2015; 115: 147–152.
- Pulitano P, Franco V, Mecarelli O, *et al.* Effects of eslicarbazepine acetate on lipid profile and sodium levels in patients with epilepsy. *Seizure* 2017; 53: 1–3.
- 22. Brodie MJ. Sodium channel blockers in the treatment of epilepsy. *CNS Drugs* 2017; 31: 527–534.
- Galiana GL, Gauthier AC and Mattson RH. Eslicarbazepine acetate: a new improvement on a classic drug family for the treatment of partialonset seizures. *Drugs RD* 2017; 17: 329–339.
- Mintzer S, Constantino T, Gidal B, et al. Markers of bone and lipid metabolism with eslicarbazepine acetate monotherapy. *Epilepsy Res* 2019; 158: 106216.
- Rocamora R, Peltola J, Assenza G, et al. Safety, tolerability and effectiveness of transition to eslicarbazepine acetate from carbamazepine or oxcarbazepine in clinical practice. Seizure 2020; 75: 121–128.

- Villanueva V, Ojeda J, Rocamora RA, et al. EPICON consensus: recommendations for proper management of switching to eslicarbazepine acetate in epilepsy. *Neurología* 2018; 33: 290–300.
- 27. Trinka E, Rocamora R, Chaves J, *et al.* Longterm efficacy and safety of eslicarbazepine acetate monotherapy for adults with newly diagnosed focal epilepsy: an open-label extension study. *Epilepsia* 2020; 61: 2129–2141.
- 28. Bullman J, Nicholls A, Van Landingham K, *et al.* Effects of lamotrigine and phenytoin on the pharmacokinetics of atorvastatin in healthy volunteers. *Epilepsia* 2011; 52: 1351–1358.
- 29. Almeida L and Soares-da-Silva P. Eslicarbazepine acetate (BIA 2-093). *Neurother* 2007; 4: 88–96.
- Hainzl D, Parada A and Soares-da-Silva P. Metabolism of two new antiepileptic drugs and their principal metabolites S(+)- and R(-)-10,11dihydro-10-hydroxy carbamazepine. *Epilepsy Res* 2001; 44: 197–206.
- Hebeisen S, Pires N, Loureiro AI, et al. Eslicarbazepine and the enhancement of slow inactivation of voltage-gated sodium channels: a comparison with carbamazepine, oxcarbazepine and lacosamide. *Neuropharmacology* 2015; 89: 122–135.
- Bonifácio MJ, Sheridan RD, Parada A, et al. Interaction of the novel anticonvulsant, BIA 2-093, with voltage-gated sodium channels: comparison with carbamazepine. *Epilepsia* 2001; 42: 600–608.
- Nebert DW and Russell DW. Clinical importance of the cytochromes P450. *Lancet* 2002; 360: 1155–1162.
- DeBose-Boyd RA. Feedback regulation of cholesterol synthesis: sterol-accelerated ubiquitination and degradation of HMG CoA reductase. *Cell Res* 2008; 18: 609–621.
- 35. Rochtus AM, Herijgers D, Jansen K, *et al.* Antiseizure medications and thyroid hormone homeostasis: literature review and practical recommendations. *Epilepsia* 2022; 63: 259–270.
- Attilakos A, Garoufi A, Voudris K, *et al.* Thyroid dysfunction associated with increased low-density lipoprotein cholesterol in epileptic children treated with carbamazepine monotherapy: a causal relationship? *Eur J Paediatr Neurol* 2007; 11: 358–361.
- 37. Mintzer S, Dimova S, Zhang Y, *et al.* Effects of lacosamide and carbamazepine on lipids in a randomized trial. *Epilepsia* 2020; 61: 2696–2704.

- Mäkinen J, Rainesalo S, Raitanen J, et al. Discontinuation of carbamazepine due to concerns of long-term consequences of enzyme induction. *Epilepsia Open* 2018; 3: 340–347.
- Peltola J, Holtkamp M, Rocamora R, et al. Practical guidance and considerations for transitioning patients from oxcarbazepine or carbamazepine to eslicarbazepine acetate–Expert opinion. Epilepsy Behav 2015; 50: 46–49.
- 40. Villanueva V, Serratosa JM, Guillamón E, *et al.* Long-term safety and efficacy of eslicarbazepine acetate in patients with focal seizures: results of the 1-year ESLIBASE retrospective study. *Epilepsy Res* 2014; 108: 1243–1252.
- 41. Villanueva V, Bermejo P, Montoya J, et al. EARLY-ESLI study: long-term experience with eslicarbazepine acetate after first monotherapy failure. *Acta Neurol Scand* 2017; 136: 254–264.

Visit Sage journals online journals.sagepub.com/ home/tan

Sage journals